Title: Validation of the Turkish version of the Centre for Epidemiologic Studies Depression Scale (CES-D) in patients with Type 2 diabetes mellitus.

Authors:

Vicky Lehmann (v.Lehmann@uvt.nl)
Ceylan Makine (ceylan_makine@hotmail.com)
Çagatay Karşıdağ (ckarsidag@yahoo.com)
Pınar Kadıoğlu (kadioglup@yahoo.com)
Kubilay Karşıdağ (karsidag@istanbul.edu.tr)
François Pouwer (F.Pouwer@uvt.nl)

Version: 3 Date: 9 July 2011

Author's response to reviews: see over
Cover letter:
format changes for the manuscript 1644923667529921
Title: Validation of the Turkish version of the Centre for Epidemiologic Studies Depression Scale (CES-D) in patients with Type 2 diabetes mellitus.

Major revisions:
#1 Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order. The preferred style is 'First name Initial Last name' (e.g. Joe F Bloggs)
Response: We think that we have provided exactly the same authors list (names and order), but think that due to some special signs (such as ş, ğ), they were not correctly recognized in the system!? We would like to encourage the editors, to use the author list as provided on our title page.

#2 Authors' contributions: Please use initials instead of full name.
Authors' contributions: Please indicate that all authors read and approved the final manuscript.
Response: both aspects have been changed on page 14

Minor revisions:
#1 Highlighting/tracking: Please remove all highlighting/tracking from the manuscript.
Response: done

#2 Authors¿ contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points.
Response: We hope that we have changed things according to the wishes of the editor.

All authors read and approved the final manuscript. VL had full access to all the data in the study and takes responsibility for the integrity of the data and the accuracy of the data analysis. The study concept and design was initiated and implemented by VL, CM, and FP. Data was acquired by CM and FP, and supported by administrative, technical, or material help of CK, PK, and KK. Statistical analysis, interpretation of results, and drafting of the manuscript was conducted by VL and supervised by FP. CM, CK, PK, KK, and FP critically revised the manuscript for important intellectual content.

#3 Tables: Please remove visible vertical lines from your tables.
Response: done

#4 Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Response: We appreciate this opportunity.